Date: February 5-8, 2001 CSI 2131 Page: 1 Profs. Moura and Japkowicz Tutorial 5

In this tutorial, the teaching assistants will discuss the various modules in the program for Huffman encoding and decoding.

The program can be found at :

http://www.site.uottawa.ca/ lucia/courses/2131/programs/A2startuphuffman/

Programs to be discussed :

- README.txt
- comparable.cpp
- comparable.h
- encoded.bin
- freq
- huffman.cpp
- huffmancharacter.cpp
- huffmancharacter.h
- huffmandecoder.cpp
- huffmandecoder.h
- huffmanencoder.cpp
- huffmanencoder.h
- huffmanobject.cpp
- huffmanobject.h
- huffmantree.cpp
- huffmantree.h
- ibitstream.cpp
- ibitstream.h
- obitstream.cpp
- obitstream.h
- priorityqueue.cpp
- priorityqueue.h
- sort.cpp
- sort.h